Author's response to reviews

Title: Computerized prediction of intensive care unit discharge after cardiac surgery: development and validation of a Gaussian processes model

Authors:

Geert Meyfroidt (geert.meyfroidt@uzleuven.be)
Fabian Güiza (fabian.guiza@med.kuleuven.be)
Dominiek Cottem (dominiek.cottem@uzleuven.be)
Wilfried De Becker (wilfried.debecker@uzleuven.be)
Kristien Van Loon (kristien.vanloon@biw.kuleuven.be)
Jean-Marie Aerts (jean-marie.aerts@biw.kuleuven.be)
Daniël Berckmans (daniel.berckmans@biw.kuleuven.be)
Jan Ramon (jan.ramon@cs.kuleuven.be)
Maurice Bruynooghe (maurice.bruynooghe@cs.kuleuven.be)
Greet Van den Berghe (greet.vandenberghe@med.kuleuven.be)

Version: 5 Date: 21 October 2011

Author's response to reviews:

Formatting changes to be made:

-----------------------------

Major revisions (we require the author to make these changes)

REPLY: done, as requested

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures
REPLY: done, to my knowledge. I refer to my previous reply.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
REPLY: done, as requested